Author’s response to reviews

Title: Guide to Bangladeshi eHealth and mHealth initiatives: A scoping study

Authors:

Tanvir Ahmed (tanvir@icddrb.org)
Henry Lucas (h.lucas@ids.ac.uk)
Azfar S Khan (azfar@icddrb.org)
Rubana Islam (irubana@icddrb.org)
Abbas U Bhuiya (abbas@icddrb.org)
Mohammad Iqbal (migbal@icddrb.org)

Version: 2 Date: 15 May 2013

Author’s response to reviews:

There was two mistakes in typing. In the title of the paper there should be 'eHealth' instead of 'ehealth' and 'mHealth' instead of 'mhealth.'